Author’s response to reviews

Title: The potential impact of Brexit and Immigration policies on the GP workforce in England: a cross sectional observational study of GP qualification region and the characteristics of the areas and population they served in September 2016

Authors:

Aneez Esmail (aneez.esmail@manchester.ac.uk)

Maria Panagioti (maria.panagioti@manchester.ac.uk)

Evangelos Kontopantelis (e.kontopantelis@manchester.ac.uk)

Version: 1 Date: 28 Sep 2017

Author’s response to reviews:

We would like to thank the two reviewers for their encouraging words and careful reading of the paper. We have now corrected the typo and made all requested changes for the paper to comply with the journal requirements.